Author's response to reviews

Title: Reliability and validity of the Brief Illness Perception Questionnaire for use in acute low back pain patients.

Authors:

Joannes M Hallegraeff (jmhallegraeff@online.nl)
Cees P van der Schans (c.p.van.der.schans@pl.hanze.nl)
Wim P Krijnen (w.p.krijnen@pl.hanze.nl)
Mathieu HG de Greef (m.h.g.de.greef@rug.nl)

Version: 3 Date: 13 June 2012

Author's response to reviews:

This is the new cover letter:


Subject: (Re) Submission of a manuscript.

Dear editor(s) of BMC Musculoskeletal Disorders,

With this writing, I would like to propose to you the resubmission of the manuscript “Reliability and validity of the Brief Illness Perception Questionnaire for use in acute low back pain patients”, for publication in BMC Musculoskeletal Disorders. All contributing authors have given their consent for submission of the manuscript.

Furthermore, with this resubmission I would like to undertake that the above mentioned manuscript has not been published elsewhere, accepted for publication elsewhere or under editorial review for publication elsewhere; and that my Institute’s (Research Group in Health Care and Nursing – Hanze University, Applied Sciences, Groningen, the Netherlands) representative is fully aware of this submission.

The study design is a non-experimental, cross-sectional study. All participants received usual standard physical therapy, which were carried out according to guidelines for ANLBP without controlling or manipulating participants or conditions. The main goal of the article is to describe the psychometric qualities of the Brief Illness Perception Questionnaire (Brief IPQ) in primary care physical therapy, specifically in acute nonspecific low back pain patients. However measurement of patients' perception of musculoskeletal disorders is increasing in daily practice, measurement instruments for those purposes have not been researched until now. It is therefore that this study is a novelty in this field of interest and relevant for daily clinical practice in primary care and for evidence
based research. Studies of patients’ perceptions in musculoskeletal disorders are rare; therefore this study is of great importance for evidence search and contributes to the field of nonspecific low back pain. Knowledge of recognizing this cognition is essential.

Our study showed adequate reliability and validity, however the Brief IPQ is not suitable for measuring changes in low back pain over time.

BMC Musculoskeletal Disorders is an open access Journal and covers all aspects of musculoskeletal disorders, that is the reason for submission this study to your Journal.

The Research Project was not sponsored.

I hope you will consider the manuscript for reviewing. If you have any questions, please don’t hesitate to contact me.

Sincerely,

Hank Hallegraeff.

The following changes have been made.

Abstract:
- In line 6 we inserted: “common musculoskeletal disorders like” to emphasize that new evidence is presented in musculoskeletal disorders.

- In line 7 the phrase “Patients’ perceptions vary across different populations and affect coping styles”, is added, relating the main reason to execute this study.

- Line 13: “in Dutch primary care” is added.

- In line 22 “these correlation” is inserted.

- Addition to line 26 “compared with the mental health subscale of the Short Form 36 Health Survey. However, limits of agreement are large.”

Manuscript:

- Line 54, 55, 56. The following sentence is inserted: “All participants received usual standard physical therapy, which were carried out according to guidelines for ANLBP without controlling or manipulating participants or conditions.”

- In line 57 the addition “directives given” is inserted.

- Line 60, 61, and 62 the section Design is added: “Design. A non-experimental cross-sectional study design was performed, with two measurement moments.”

- In line 188 “perceptions” is changed into “perception”.


“Authors’ contributions and competing interests” have changed and expanded.

Contact:

J.M. Hallegraeff, PT, MSc
Research Group in Health Care and Nursing – Hanze University, Applied Sciences, Groningen, the Netherlands,
Master Education Manual Therapy SOMT, Amersfoort, the Netherlands,
Editor Dutch Journal Manipulative Therapy, the Netherlands.
Primary Care Medical Care Center, Groningen, the Netherlands.
jmhallegraeff@online.nl +31655955742.